Reassuring pregnancy outcomes in women with mild *COL4A3-5*-related disease (Alport syndrome) and genetic type of disease can aid personalized counseling



OPEN

Margriet E. Gosselink^{1,2}, Rozemarijn Snoek^{1,2}, Agne Cerkauskaite-Kerpauskiene³, Sophie P.J. van Bakel^{1,2}, Renee Vollenberg^{1,2}, Henk Groen⁴, Rimante Cerkauskiene⁵, Marius Miglinas³, Rossella Attini⁶, Kálmán Tory⁷, Kathleen J. Claes⁸, Kristel van Calsteren⁹, Aude Servais¹⁰, Margriet F.C. de Jong¹¹, Valentine Gillion¹², Liffert Vogt¹³, Antonio Mastrangelo¹⁴, Monica Furlano¹⁵, Roser Torra¹⁵, Kate Bramham¹⁶, Kate Wiles¹⁷, Elizabeth R. Ralston¹⁶, Matthew Hall¹⁸, Lisa Liu¹⁹, Michelle A. Hladunewich¹⁹, A. Titia Lely^{2,20} and Albertien M. van Eerde^{1,20}; on behalf of the ALPART working group

¹Department of Genetics, University Medical Center Utrecht, Utrecht, the Netherlands; ²Department of Obstetrics, University Medical Center Utrecht, Utrecht, the Netherlands; ³Clinic of Gastroenterology, Nephro-Urology and Surgery, Institute of Clinical Medicine, Faculty of Medicine, Vilnius University, Vilnius, Lithuania; ⁴Department of Epidemiology, University of Groningen, University Medical Center Groningen, Groningen, the Netherlands; ⁵Clinic of Children's Diseases, Institute of Clinical Medicine, Faculty of Medicine, Vilnius University, Vilnius, Lithuania; ⁶Department of Obstetrics and Gynecology SC2U, Città della Salute e della Scienza, Sant'Anna Hospital, Turin, Italy; ⁷MTA-SE Lendulet Nephrogenetic Laboratory, Pediatric Center, Semmelweis University, Budapest, Hungary; ⁸Department of Nephrology, University Hospital Leuven, Leuven, Belgium; ⁹Department of Obstetrics and Gynaecology, University Hospital Leuven, Leuven, Belgium; ¹⁰Department of Nephrology and Transplantation, Necker Enfants Maladies University Hospital, Assistance Publique Hôpitaux de Paris, Paris, France; ¹¹Division of Nephrology, Department of Internal Medicine, University Medical Center Groningen, Groningen, the Netherlands; ¹²Department of Nephrology, Cliniques Universitaires Saint-Luc (Université Catholique de Louvain), Brussels, Belgium; ¹³Section Nephroloay, Department of Internal Medicine, Amsterdam University Medical Centers, University of Amsterdam, Amsterdam Cardiovascular Sciences, Amsterdam, the Netherlands; ¹⁴Pediatric Nephrology, Dialysis, and Transplant Unit, Fondazione IRCCS Ca⁷ Granda Ospedale Maggiore Policlinico, Milan, Italy; ¹⁵Department of Nephrology, Inherited Kidney Diseases, Fundació Puigvert, Institut d'Investigacions Biomèdiques Sant Pau Universitat Autònoma de Barcelona, RICORS2040 (Kidney Disease), Barcelona, Spain; ¹⁶Department of Women and Children's Health, Kina's Colleae London, London, UK; ¹⁷Department of Women and Children, Barts National Health Service Trust and Queen Mary University of London, London, UK; ¹⁸Department of Nephrology, Nottingham University Hospitals, Nottingham, UK; and ¹⁹Division of Nephrology, Department of Medicine, Sunnybrook Health Sciences Centre, Temerty Faculty of Medicine, Toronto, Ontario, Canada

Individualized pre-pregnancy counseling and antenatal care for women with chronic kidney disease (CKD) require disease-specific data. Here, we investigated pregnancy outcomes and long-term kidney function in women with *COL4A3-5* related disease (Alport Syndrome, (AS)) in a large multicenter cohort. The ALPART-network (mAternaL and fetal PregnAncy outcomes of women with AlpoRT syndrome), an international collaboration of 17 centers, retrospectively investigated *COL4A3-5* related disease pregnancies after the 20th week. Outcomes were stratified per inheritance pattern (X-Linked AS (XLAS)), Autosomal Dominant AS (ADAS), or Autosomal Recessive AS (ARAS)). The influence of pregnancy on estimated glomerular filtration rate (eGFR)-slope was assessed in 192 pregnancies encompassing 116 women (121 with XLAS, 47 with ADAS, and 12 with ARAS). Median eGFR pre-pregnancy was over 90ml/min/1.73m². Neonatal outcomes were favorable: 100% live births, median gestational age 39.0 weeks and mean birth weight 3135 grams. Gestational hypertension occurred during 23% of pregnancies (reference: 'general' CKD G1-G2 pregnancies incidence is 4-20%) and preeclampsia in 20%. The mean eGFR declined after pregnancy but remained within normal range (over 90ml/min/1.73m²). Pregnancy did not significantly affect eGFR-slope (pre-pregnancy $\beta = -1.030$, post-pregnancy β = -1.349). ARAS-pregnancies demonstrated less favorable outcomes (early preterm birth incidence 3/11 (27%)). ARAS was a significant independent predictor for lower birth weight and shorter duration of pregnancy, next to the classic predictors (pre-pregnancy kidney function, proteinuria, and chronic hypertension) though missing proteinuria values and the small ARAS-sample hindered analysis. This is the largest

Correspondence: Margriet E. Gosselink, Departments of Clinical Genetics and Obstetrics and Gynaecology, University Medical Centre Utrecht location Wilhelmina, Children's Hospital, PO Box 85090, 3508 AB Utrecht, the Netherlands. E-mail: m.e.gosselink-4@umcutrecht.nl

²⁰ATL and AMvE share last authorship.

Received 25 February 2023; revised 22 December 2023; accepted 8 January 2024; published online 19 February 2024

study to date on AS and pregnancy with reassuring results for mild AS, though inheritance patterns could be considered in counseling next to classic risk factors. Thus, our findings support personalized reproductive care and highlight the importance of investigating kidney disease-specific pregnancy outcomes.

Kidney International (2024) **105,** 1088–1099; https://doi.org/10.1016/ j.kint.2024.01.034

KEYWORDS: Alport syndrome; COL4A3-5-related disease; long-term kidney function; pregnancy outcomes; pre-pregnancy counseling

Copyright © 2024, International Society of Nephrology. Published by Elsevier Inc. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Lay Summary

Our study, the largest in its kind, investigated 192 pregnancies in 116 women with Alport syndrome (AS). AS is a rare genetic kidney disease and can cause pregnancy complications. Pre-pregnancy kidney function, hypertension, and urinary protein loss are risk factors for pregnancy complications in women with kidney disease in general. Data on disease-specific outcomes are needed. Together with 17 medical centers worldwide, we studied pregnancy outcomes and long-term kidney health in overall mild AS cases. Our findings are positive: babies were born after normal pregnancy duration with normal birth weight (average, 39 weeks; 3135 g). Kidney function declined after pregnancy within normal range, and pregnancy did not worsen rate of decline. However, women with a severe form of AS (autosomal recessive AS) experienced less favorable outcomes. Our results offer reassurance and help physicians tailor pregnancy care for women with AS. Considering the inheritance type of ASalongside traditional risk factors—could help predict complications.

hronic kidney disease (CKD) affects $\approx 3\%$ of pregnancies, conveying higher risks of complications, such as preeclampsia, fetal growth restriction, and preterm birth.¹⁻⁴ Controversy exists about the exact impact of pregnancy on long-term kidney function in different CKD stages. Recent studies in the kidney transplant population⁵ and advanced CKD population (stage 3–5)⁶ showed a stepwise decline in estimated glomerular filtration rate (eGFR) after pregnancy, but no significant influence of pregnancy on eGFR slope. Most studies in pregnant women with CKD are based on heterogeneous cohorts, without differentiating outcomes according to cause of kidney disease.^{3,7,8} Their different reported pregnancy outcomes might be explained by the modifying effect of cause. Current pre-pregnancy counseling is based on pre-pregnancy CKD category, chronic hypertension, and proteinuria.²⁻ 4,9,10 Adding data on cause-based pregnancy outcomes could improve individualized counseling.

One of the causes lacking data on pregnancy outcomes is Alport syndrome (AS), also known as COL4A3-5-related disease. This is a hereditary kidney disease, caused by pathogenic variants in the collagen type IV genes (COL4A3, COL4A4, and COL4A5), leading to defect basement membranes of the ears, eyes, and kidneys. In the kidneys, defect glomerular basement membranes lead to hematuria, proteinuria, progressive loss of kidney function, and kidney failure.^{11,12} Different inheritance patterns for COL4A3-5-related disease exist: X-linked AS (XLAS; COL4A5 gene; incidence, 65%-80%), autosomal dominant AS (ADAS; COL4A3 or COL4A4 heterozygotes; 5%-20%), autosomal recessive AS (ARAS; COL4A3 or COL4A4 homozygotes or compound heterozygotes; 15%), or digenic.^{13–15} There is growing consensus on classifying COL4A3 or COL4A4 heterozygotes as having ADAS instead of being carriers of AS, because they may develop proteinuria, hypertension, or kidney impairment themselves as well.^{12,15–17} Despite high variability in phenotype, in general, ARAS shows more severe phenotypes in women than XLAS and ADAS, with often considerable proteinuria during childhood, onset of endstage kidney disease before 30 years of age, and more severe vascular and kidney damage. ADAS can generally be seen as the mildest form of AS, with the lowest risk of end-stage kidney disease.^{11,12,14,15} Diagnosis of AS can be established by either genetic testing or kidney or skin biopsy. Because of underdiagnosis and incomplete penetrance of pathogenic variants, AS diagnosis prevalence is estimated at 1:50,000 live births, whereas the predicted population prevalence of pathogenic COL4A5 variants is 1:2230 and heterozygous pathogenic COL4A3 and COL4A4 variants 1:106 in people without known kidney disease.^{14,18} These numbers suggest COL4A3-5-related disease to be 1 of the most frequent hereditary kidney diseases.

Literature on *COL4A3-5*–related disease and pregnancy is limited to \approx 30 reported cases. Reports range from having pre-pregnancy microscopic hematuria with favorable outcomes to proteinuria progressing to nephrotic range during pregnancy, hypertensive complications, and kidney replacement therapy (KRT) after pregnancy.^{19–29} Given the existing literature being limited to case reports, predictors for adverse outcomes and correlation with genetic aspects could not be assessed. To better counsel patients with *COL4A3-5*–related disease who want to conceive, more information on pregnancy outcomes is needed. Therefore, the aim of this study is to report on pregnancy outcomes in women with *COL4A3-5*– related disease and to determine the influence of pregnancy on long-term kidney function.

METHODS

Study design and participants

A retrospective cohort study was conducted within the ALPART network (mAternal and foetal PregnAncy outcomes of women with AlpoRt syndrome). In 2019, the University Medical Center Utrecht established the ALPART network, an international collaboration of 17 medical centers, to investigate Alport-specific pregnancy outcomes (see Supplementary Table S1 for participants). A survey of the European Rare Kidney Disease Reference Network Workgroup Hereditary Glomerulopathies and the international network of the research group identified ALPART collaborators. Participating sites identified pregnancies in women with COL4A3-5-related disease through medical file review. Eligible women included adults with ≥ 1 singleton pregnancy beyond 20 weeks' gestation with either (i) established genetic COL4A3-5-related disease diagnosis or (ii) clinically confirmed diagnosis. A clinically confirmed diagnosis was defined as (extra) renal symptoms (extrarenal: i.e., sensorineural hearing loss, ocular symptoms, such as lenticonus) and biopsy with thinning or irregular thickness of the glomerular basement membrane and/or skin biopsy with aberrant collagen α -chain staining. Extrarenal features were entered as stated in medical files, and separate audiometry results could not be retrieved. Genetic diagnosis was defined according to the American College of Medical Genetics framework for variant classification: pathogenic (class 5) or likely pathogenic (class 4) variants were considered a genetically proven diagnosis.³⁰ Patients with multiple variants of which ≥ 1 (likely) pathogenic and further variants of unknown significance were also included. Clinical and genetic diagnoses were reviewed by 2 research team members (MEG, AMvE), to establish and ascertain AS diagnosis, inheritance pattern, and inclusion. Data were collected from November 14, 2019, until December 31, 2022. This study was approved by the University Medical Center Utrecht medical ethics committee (19-238/C) and in participating centers according to local regulations.

Data collection and definitions

Data were pseudonymized and retrospectively collected by a dedicated research team member at each hospital. Data were entered using electronic case report forms into the data capture tool Castor.³¹ Kidney disease characteristics, including genotype information (excluding information on base pair alterations to preserve anonymity) and maternal and neonatal pregnancy outcomes, were collected. Inheritance patterns were grouped according to the following combination of genetic and clinical criteria. XLAS: (likely) pathogenic variant in COL4A5 gene (including variants of unknown significance/missing American College of Medical Genetics status with biopsy-proven diagnosis); ADAS: heterozygous (likely) pathogenic variant in COL4A3 or COL4A4 gene (including variants of unknown significance/missing American College of Medical Genetics status with biopsy-proven diagnosis); and ARAS: having at least 2 homozygous/compound heterozygous maternal and paternal (likely) pathogenic variants in COL4A3 or COL4A4 gene. In case family segregation or pathogenicity was unavailable, severe phenotypes in combination with the compound heterozygous genotype were accepted as ARAS.

eGFR was calculated using the 2021 Chronic Kidney Disease Epidemiology Collaboration creatinine equation without ethnicity adjustment.³² Routinely collected serum creatinine concentrations (SCr) and date of measurement were collected at up to 9 time points: at diagnosis, 1 to 5 years pre-pregnancy, most recent pre-pregnancy, during each trimester of pregnancy, 3 months postpartum, 1 year postpartum, and most recent to date where available. Local investigators selected only representative values for data entry. In case of need for KRT, no SCr values were entered after. Chronic hypertension was defined as systolic blood pressure \geq 140 mm Hg and/or diastolic blood pressure \geq 90 mm Hg before pregnancy,³³ use of antihypertensive medication at conception (isolated use of reninangiotensin blockade excluded), and diastolic blood pressure >85 mm Hg at <16 weeks' gestation.³⁴ An extra category, "unlikely chronic hypertension," was added, for women in whom prepregnancy data were not retrievable (see Supplementary Methods for definition). Gestational hypertension was defined similar to chronic hypertension, only occurring at >16 weeks' gestation when no chronic hypertension previously existed. Hypertension during pregnancy was defined as gestational hypertension, only being assessed independently from the occurrence of chronic hypertension. This parameter was added because chronic hypertension status was not known for all pregnancies, and in these cases the distinction between truly new-onset (gestational) or chronic hypertension could not be determined with certainty. Proteinuria was defined as proteinto-creatinine ratio >15 mg/mmol, protein excretion rate >150 mg/ 24 h, protein in urine portion >140 mg/L, positive urine dipstick, use of renin-angiotensin blockade, or if stated in medical file.³⁵ Proteinuria during pregnancy was defined similarly, only occurring during pregnancy. Nephrotic-range proteinuria was defined as protein-to-creatinine ratio >350 mg/mmol or protein excretion rate >3500 mg/24 h.^{35,36} Preeclampsia was defined by the attending physician during pregnancy, by the presence of gestational hypertension and proteinuria (>300 mg/mmol). This classification could not be uniformly defined retrospectively because of missing values. Small for gestational age was defined as birth weight below the 10th percentile on the World Health Organization Fetal Growth Charts.³⁷ See Supplementary Methods for additional definitions. Pregnancies after kidney transplantation were excluded. Experienced health care professionals (RS and MG) checked data rigorously and asked collaborators to provide missing values and clarifications.

Study end points

Maternal primary outcome was gestational hypertension. Fetal primary outcomes were live birth rate, duration of pregnancy, and neonatal birth weight. Secondary outcomes were long-term kidney function after pregnancy and the association of X-linked, autosomal dominant and autosomal recessive inheritance patterns and pregnancy outcomes.

Statistical analysis

As in previously published pregnancy cohorts, women were allowed to contribute ≥ 1 pregnancy, and analyses were performed at pregnancy level.³⁸ Normality of data was determined graphically by using histograms. Normally distributed continuous variables were reported as mean (SD). Skewed data were reported as median (interquartile range [IQR]). Considering the nonindependence of multiple pregnancies in 1 woman, data had a multilevel structure and were analyzed by multilevel analysis. Baseline characteristics and pregnancy outcomes were analyzed for the total cohort and stratified by inheritance pattern. Association between inheritance pattern and pregnancy outcomes was tested in multilevel prediction analysis. In this analysis, ADAS inheritance pattern was used as reference category.

The effect of pregnancy on long-term kidney function was investigated using generalized estimated equations analysis: an established method for multilevel analysis.³⁹ This way, eGFR trajectory could be analyzed despite varying numbers of measurements per pregnancy and clustered data from women with multiple pregnancies. As subject variable, the patient identifier was used. As within-subject variables, number of pregnancy (1, 2, 3, or 4) was used as well as measuring point of eGFR measurement (1 of 6 possible time points in data collection). The effect of time on eGFR (eGFR slope) was established by maternal age at eGFR measurement and was added to the model as a continuous covariate. Dates of eGFR measurements were divided into 4 "pregnancy intervals": interval 0 (before first pregnancy), interval 1 (between first and second

Table 1 | Baseline characteristics of the total study cohort

Variables		Total coh	ort (n = 192)		
General characteristics					
Caucasian ethnicity	157/180 (87)				
Nullipara	102/177 (58)				
Age at conception, yr		2	9.3 (6.9)		
BMI at conception, kg/m ²		2	2.5 (4.8)		
Decennium delivery					
1970–1980		6/1	190 (3)		
1980–1990		16/1	190 (8)		
1990-2000	30/1	190 (16)			
2000-2010		45/1	190 (24)		
Kidney disease_specific chara	ctorictics	93/1	190 (49)		
Pre-pregnancy eGER ^e ml/mi	n per 173 m ²	1	14 (31 5)		
Pre-pregnancy serum creatin	line, ^e umol/l		64 (19)		
CKD category, most recent p	pre-pregnancy				
G1		102/1	31 (78)		
G2		23/1	31 (18)		
G3 ^a		4/1	31 (3)		
G3 ^b		1/1	31 (1)		
G4		1/1	31 (1)		
G5			0		
Chronic hypertension pre-pre	egnancy	21/1	74 (12)		
No		118/1	74 (68)		
Unlikely		35/1	74 (20)		
Pre-pregnancy proteinuria		89/1	122 (73)		
Homaturia		171/1	184 (03)		
Extrarenal features		171/104 (93)			
Hearing loss		45/166 (27)			
Ocular symptoms		30/136 (22)			
Diagnosis			,		
Genetic testing performed		187/1	187/192 (97)		
Established genetic diagno	osis (LP/P)	166/187 (89)			
VUS		7/1	87 (4)		
ACMG unknown		14/1	87 (7)		
Inclusion cohort based on					
Genetic diagnosis (LP/P)		166/192 (86)			
Clinical biopsy-proven diag	gnosis	26/192 (14)			
Kidney biopsy		25/26 (96)			
yi Asb		171/1	(20 (4) 180 (67)		
		121/1	180 (07)		
ABAS ^d		12/1	180 (20) 180 (7)		
Different variants nor cone	COL 445	<u> </u>	<u> </u>		
	COL4AS		COL4AS		
Missense					
Collagenous domain	52/107 (49)	4/18 (22)	36/39 (92)		
Glycine substitution	49/52	4/4	33/36		
Noncollagenous domain	1/10/(1)	0	1/39 (3)		
Deletion or duplication	24/107 (22) 15/107 (14)	7/10 (39)	1/39 (3)		
Premature stop codon	7/107 (14)	2/18 (20) 2/18 (11)	0		
Frameshift	5/107 (5)	0	0		
Other 3/107 (3)		õ	õ		
Prenatal diaanosis	2, . 2, (2)	5	5		
Preimplantation genetic	3/67 (4)				
testing performed					
Prenatal genetic testing/	17/159 (11)				
prenatal diagnosis					
performed					

ACMG, American College of Medical Genetics; ADAS, autosomal dominant Alport syndrome; ARAS, autosomal recessive Alport syndrome; BMI, body mass index; CKD, chronic kidney disease; eGFR, estimated glomerular filtration rate; IQR, interquartile range; LP/P, likely pathogenic/pathogenic; VUS, variant of uncertain significance; XLAS, X-linked Alport syndrome.

^aDefinition of proteinuria (both pre-pregnancy and during pregnancy), if meeting ≥ 1 of the following criteria: protein-to-creatinine (continued)

(continued) ratio >15 mg/mmol; protein excretion rate >150 mg/24 h; protein in urine portion >140 mg/L; positive dipstick proteinuria; use of renin-angiotensin blockade; or stated in the medical file.

^bOf which 1 VUS and 11 ACMG unknown.

^cOf which 2 VUS and 1 ACMG unknown.

^dOf which n = 2 pregnancies with 1 variant ACMG 4: likely pathogenic and 1 variant ACMG 3: VUS; and n = 3 pregnancies with unknown segregation or pathogenicity. These pregnancies were classified ARAS because of the combination of likely compound heterozygosity and severe clinical phenotype (i.e., extrarenal features and kidney failure at young age, <50 years). ^eMost recent before pregnancy.

Data are presented as median (IQR) unless stated otherwise. Data on genotype are presented on a pregnancy level and not on patient level. Not all baseline characteristics were available for all pregnancies. For categorical variables, incidences are shown as numerator/denominator; for continuous variables, the number of missing cases is described here: age at conception: n = 4; BMI at conception: n = 52 (27%); eGFR prepregnancy: n = 61 (32%); serum creatinine concentration pre-pregnancy: n = 61 (32%).

pregnancy), interval 2 (between second and third pregnancy), and interval 3 (after third pregnancy). The reference category was interval 0. The interaction term "pregnancy interval × age at eGFR measurement (time)" was added to the model to investigate whether pregnancy accelerates eGFR decline over time. An unstructured correlation matrix was used. eGFR measurements during pregnancy, after KRT, and at age <18 years were excluded.

Multilevel analyses were performed to (i) identify other factors' association with eGFR trajectory and (ii) establish the influence of pregnancy on eGFR slope, adjusted for these factors. Previously established associations with eGFR, such as pre-pregnancy kidney function, body mass index, chronic hypertension, and proteinuria, were added to analyses.^{35,40–43} The dichotomous variable "after first pregnancy" appoints timing of eGFR measurement before/after first pregnancy (including pregnancy interval 1–3). Furthermore, effect of inheritance pattern and decennium of pregnancy on eGFR slope were investigated. P < 0.05 was considered significant. Analyses were performed using IBM SPSS Statistics version 27.0 (SPSS Inc.) and GraphPad Prism version 8.4.1 (Graph Pad Software Inc.).

Comparison to published cohorts

COL4A3-5–related disease pregnancy outcomes were compared with earlier published general CKD cohorts and the general Dutch population.^{4,44,45}

RESULTS

Baseline

In total, 192 pregnancies in 116 women were included. Four pregnancies after kidney transplantation were excluded (Supplementary Figure S1). Table 1 shows baseline characteristics. Median age at pregnancy was 29.3 years (IQR, 6.9 years). Median pre-pregnancy eGFR was >90 ml/min per 1.73 m² (median, 114; IQR, 31.5). Median time between pregnancy and pre-pregnancy eGFR measurement was 0.62 years (IQR, 0.9 years). Most pregnancies occurred in women with pre-pregnancy CKD stage G1 to G2 (78% and 18%, respectively). Chronic hypertension occurred in 12%, and prepregnancy proteinuria occurred in 73%, of pregnancies (36% missing values). For most pregnancies, a genetic diagnosis had been established (86% vs. 14% clinical). In 48 of 145 (33%) of pregnancies, COL4A3-5-related disease was diagnosed before pregnancy. Nevertheless, in 106 of 131 (81%) pregnancies, clinical signs of AS were present pre-pregnancy already (see Supplementary Methods for definition). Most pregnancies (65%) had genetic variant(s) in COL4A5, 23% in COL4A3, and

10% in *COL4A4*. XLAS occurred in 121 pregnancies (67%), ADAS in 47 pregnancies (26%), and ARAS in 12 pregnancies (7%). Most variants in *COL4A5* and *COL4A3* were missense variants in the collagenous domain, of which were mostly glycine substitutions. In *COL4A4*, splice site variants occurred most frequently.

Complex suspected digenic inheritance

Three pregnancies showed complex suspected digenic inheritance patterns of both variants of unknown significance and (likely) pathogenic variants in *COL4A3* and *COL4A4* and were not considered for inheritance pattern analyses. For further description, see Supplementary Table S2.

Pregnancy outcomes, maternal

Table 2 shows pregnancy outcomes. Gestational hypertension occurred in 33 of 143 (23%) of pregnancies. New-onset or doubling of proteinuria occurred in 54 of 71 (76%) of pregnancies. Doubled proteinuria levels restored postpartum in 31 of 42 pregnancies (74%). Nephrotic-range proteinuria occurred in 19 of 55 (35%) of pregnancies. Quantifiable proteinuria values during all trimesters of pregnancy were

Table 2 P	regnancy	outcomes	of the	total	study	cohort
-------------	----------	----------	--------	-------	-------	--------

Variables	Total cohort (n = 192)
Maternal variables	
Hypertension during pregnancy	56/152 (37)
Gestational hypertension	33/143 (23)
Preeclampsia	37/182 (20)
Proteinuria during pregnancy	125/148 (85)
New onset or doubling of proteinuria during	54/71 (76),
pregnancy	missing 63%
Nephrotic-range proteinuria during pregnancy	19/55 (35),
	missing 71%
Induction of delivery	67/166 (40)
Maternal indication	49/57 (86)
Fetal indication	8/57 (14)
Cesarean section	40/182 (22)
Need for kidney replacement therapy after	23/187 (12)
pregnancy	17/107 (0)
Didiysis Kida ay transmission	1//18/ (9)
Kioney transplantation	10/187 (9)
Neonatal variables	
Live birth	190/190 (100)
Gestational age, median (IQR), wk	39.0 (2.4)
	(range, 26.3–42.0)
Preterm delivery	
<37 wk	32/189 (17)
<34 wk	13/189 (7)
Newborn sex, male	102/185 (55)
Birth weight, g	3135 (724)
SGA, <p10< td=""><td>36/184 (20)</td></p10<>	36/184 (20)
<p3< td=""><td>15/184 (8)</td></p3<>	15/184 (8)
NICU admittance within 48 h after delivery	12/172 (7)
Perinatal death (32 wk GA-7 d pp)	1/188 (0.5)

GA, gestational age; IQR, interquartile range; NICU, neonatal intensive care unit; p3, 3rd percentile; p10, 10th percentile; pp, postpartum; SGA, small for gestational age (according to the World Health Organization fetal growth calculator).³⁷

Data are presented as mean (SD) and n (%) unless stated otherwise. Not all pregnancy outcomes were available for all pregnancies. For categorical variables, incidences are shown as numerator/denominator; for continuous variables, the number of missing values is described here: gestational age at delivery, n = 3; birth weight, n = 8.

missing for 60% to 70% of pregnancies. After 23 of 187 (12%) pregnancies, there was need for KRT: 17 of 187 (9%) received dialysis, and kidney transplantation was needed after 16 of 187 (9%) of pregnancies. Median start of KRT was 13.4 years (IQR, 12.2 years) after first delivery.

Pregnancy outcomes, neonatal

Live birth rate at >20 weeks' gestational age was 100% (data for 2 pregnancies were missing). Median gestational age was 39.0 weeks (IQR, 2.4 weeks), and mean birth weight was 3135 g (SD, 724 g). A total of 67 of 166 (40%) of deliveries were induced, mostly on maternal indication (49 of 57 [86%]). Preterm birth occurred in 32 of 189 (17%), and early preterm birth (<34 weeks) occurred in 13 of 189 (7%) of pregnancies. Supplementary Table S3 shows baseline characteristics of preterm births.

Pregnancy outcomes, inheritance patterns

Supplementary Table S4 shows baseline characteristics of pregnancies stratified per inheritance pattern. Age at conception, pre-pregnancy SCr, and hearing loss differed significantly between groups. Table 3 shows pregnancy outcomes stratified per inheritance pattern. Duration of pregnancy and birth weight significantly differed between ADAS, XLAS, and ARAS pregnancies: 39.0 (SEM, 0.29), 38.5 (SEM, 0.27) and 36.0 (SEM, 0.75) weeks (P = 0.001); and 3751 (SEM, 303), 3176 (SEM, 82) and 2495 (SEM, 193) g (P <0.001), respectively. The incidence of hypertensive disorders in pregnancy was significantly different: 22% (ADAS), 38% (XLAS), and 73% (ARAS) (P = 0.036). ARAS-inheritance pattern was independently associated with lower birth weight: univariable (β , -1256; SEM, 359; P < 0.001) and multivariable (β , -1135; SEM, 379; P = 0.003) when adjusted for pre-pregnancy SCr (β , 5.10; SEM, 5.25; P = 0.331), prepregnancy proteinuria (β , -153; SEM, 81; P = 0.059), and chronic hypertension (β , -451; SEM, 263; P = 0.087). Also, ARAS-inheritance pattern was independently associated with shorter duration of pregnancy: univariable (β , -2.96; SEM, 0.80; P < 0.001) and multivariable (β , -3.31; SEM, 1.33; P =0.013), although when adjusted for pre-pregnancy proteinuria (β , -1.88; SEM, 0.42; *P* < 0.001), pre-pregnancy SCr (β , 0.01; SEM, 0.02; P = 0.687), and chronic hypertension (β , -0.16; SEM, 0.89; P = 0.855), pre-pregnancy proteinuria was a stronger predictor. Figure 1 shows adjusted marginal means of duration of pregnancy and birth weight, stratified per inheritance pattern.

Pregnancy outcomes, long-term kidney function (eGFR slope)

A total of 363 eGFR measurements were analyzed in 110 patients. Mean follow-up after first pregnancy was 4.2 years (SD, 8.88 years). Kidney function declined over time, with eGFR slope of -1.47 ml/min per 1.73 m² per year (SEM, 0.15 ml/min per 1.73 m² per year; 95% confidence interval, -1.76 to -1.18 ml/min per 1.73 m² per year). eGFR after first pregnancy was significantly lower than pre-pregnancy: 93.58 (SEM, 2.72) versus 99.97 (SEM, 2.20) ml/min per 1.73 m²

Table 3 Pregnancy outcomes	, total study cohort and	stratified per inheritance pattern
------------------------------	--------------------------	------------------------------------

Variables	ADAS (N = 47)	XLAS (N = 121)	ARAS (N = 12)	P value
Maternal variables during pregnancy				
Hypertension during pregnancy	8/36 (22)	36/96 (38)	8/11 (73)	0.036
Preeclampsia	5/46 (11)	24/114 (21)	5/12 (42)	0.074
Proteinuria during pregnancy	30/37 (81)	74/89 (83)	12/12 (100)	NA
New onset or doubling of proteinuria	15/19 (79)	33/44 (75)	5/6 (83)	0.696
Nephrotic-range proteinuria during pregnancy	3/11 (27)	11/35 (31)	4/4 (100)	0.708
Induction of delivery	14/41 (34)	40/103 (39)	8/12 (67)	0.094
Cesarean section	6/46 (13)	27/113 (24)	4/12 (33)	0.563
Neonatal outcomes				
Live birth	46/46 (100)	121/121 (100)	12/12 (100)	NA
Newborn sex, male	27/45 (60)	66/116 (57)	6/12 (50)	0.072
Gestational age, mean (SEM), wk	39.0 (0.29)	38.5 (0.27)	36.0 (0.75)	0.001
Preterm delivery				
<37 wk	5/46 (11)	17/121 (14)	5/11 (46)	0.017
<34 wk	0	8/121 (7)	3/11 (27)	NA
Birth weight, mean (SEM), g	3751 (303)	3176 (82)	2495 (193)	<0.001
Small for gestational age, <p10< td=""><td>4/46 (9)</td><td>24/117 (21)</td><td>5/11 (46)</td><td>0.075</td></p10<>	4/46 (9)	24/117 (21)	5/11 (46)	0.075
NICU admittance within 48 h after delivery	2/43 (5)	7/108 (7)	2/11 (18)	NA
Perinatal death (32 wk GA–7 d pp)	0	1/120 (1)	0	NA

ADAS, autosomal dominant Alport syndrome; ARAS, autosomal recessive Alport syndrome; GA, gestational age; NA, not applicable; NICU, neonatal intensive care unit; pp, postpartum; XLAS, X-linked Alport syndrome.

In multilevel analysis, ADAS inheritance pattern was used as reference category. Continuous variables, stratified per different inheritance pattern, are presented as marginal means (SEM). Continuous variables for the total cohort are presented as mean (SD) and number/total (percentage) unless stated otherwise. Not all pregnancy outcomes were available for all pregnancies. For categorical variables, incidences are shown as numerator/denominator; and for continuous variables, the number of missing values is described here: gestational age, n = 3; birth weight, n = 8. Bold variables are significant (P < 0.05).

(P = 0.004). eGFR after second and third pregnancy was not significantly lower (97.94 [SEM, 3.18] ml/min per 1.73 m² [P = 0.564] and 99.80 [SEM, 4.96] ml/min per 1.73 m² [P = 0.975]).

Mean eGFR decline, adjusted for time, in pregnancy interval 1 was -6.40 ml/min per 1.73 m² (SEM, 2.25 ml/min per 1.73 m²; 95% confidence interval, -10.80 to -1.99 ml/min per 1.73 m²) over 1.92 years (IQR, 2.70 years). In



Figure 1 | Marginal mean duration of pregnancy and birth weight, stratified per inheritance pattern. Marginal means are adjusted for covariates pre-pregnancy serum creatinine concentration, pre-pregnancy proteinuria, and chronic hypertension in multilevel analysis (generalized estimating equations). Autosomal recessive Alport syndrome (ARAS)–inheritance pattern was an independent predictor for lower birth weight (P = 0.003) and shorter duration of pregnancy (P = 0.013). In autosomal dominant Alport syndrome (ADAS), X-linked Alport syndrome (XLAS), and ARAS pregnancies, adjusted marginal mean birth weight was 3162 g (SEM, 238 g; n = 26), 2997 g (SEM, 154 g; n = 68), and 2027 g (SEM, 235 g; n = 7; P = 0.003), respectively. Adjusted marginal mean duration of pregnancy was 38.75 weeks (SEM, 0.60 weeks; n = 26), 38.76 weeks (SEM, 0.53 weeks; n = 71), and 35.44 weeks (SEM, 1.13 weeks; n = 7; P = 0.013), respectively.

pregnancy interval 2, mean eGFR decline was -2.04 ml/min per 1.73 m² (SEM, 3.53 ml/min per 1.73 m²; 95% confidence interval, -8.96 to 4.89 ml/min per 1.73 m²) over 3.83 years (IQR, 7.67 years). After third pregnancy, mean eGFR decline was -0.18 ml/min per 1.73 m² (SEM, 5.55 ml/min per 1.73 m²; 95% confidence interval, -11.05 to 10.70 ml/ min per 1.73 m²) over 7.33 years (IQR, 9.75 years). Figure 2 illustrates adjusted marginal means of eGFR per pregnancy interval. Despite a stepwise decline in eGFR after first pregnancy, pregnancy had no significant accelerating effect on eGFR decline (P = 0.144). The interaction term was not significant for pregnancy interval 1 (β , -0.64; SEM, 0.35; P = 0.069), pregnancy interval 2 (β , -0.061; SEM, 0.44; P =0.890), and pregnancy interval 3 (β , -0.51; SEM, 0.70; P =0.467). Therefore, pregnancy did not have an additional effect on eGFR slope (overall pre-pregnancy $\beta = -1.030$; postpregnancy [includes intervals 1, 2, and 3] $\beta = -1.349$; P = 0.386). Figure 3 shows the adjusted estimated marginal means of eGFR visualized in years of follow-up before and after first pregnancy.

Other predictors that affect eGFR

To determine the effect and to adjust for it, other possible predictors for eGFR were analyzed (Table 4). For this analysis, the variable "after first pregnancy" designated all eGFR measurements before and after first pregnancy. The variables after first pregnancy, pre-pregnancy SCr, proteinuria prepregnancy, chronic hypertension, and XLAS and ARAS inheritance patterns were significantly associated with lower eGFR values in multilevel analysis adjusted for age at eGFR measurement. These variables were added to the multivariable model, including the interaction term "after first pregnancy × age at eGFR measurement," to see if eGFR slope differed per pregnancy interval when adjusted for these factors. eGFR slope after first pregnancy was not different, adjusted for the above mentioned factors (P = 0.566).

DISCUSSION

Our study has 4 key findings. First, this is the largest cohort of pregnancies in women with *COL4A3-5*–related disease to date. We show that overall neonatal pregnancy outcomes



Figure 2 | Adjusted marginal means of estimated glomerular filtration rate (eGFR) before and after pregnancies, stratified per pregnancy interval (n = 110 patients). The green line represents the eGFR of 90 ml/min per 1.73 m². eGFR decline per pregnancy interval is shown in marginal means (SEM). Annual eGFR decline in the cohort was -1.47 ml/min per 1.73 m² (SEM, 0.15 ml/min per 1.73 m²). In this model, "age at eGFR measurement" was used as a continuous covariate, and "pregnancy interval" was used as a categorical factor. Error bars represent SEM. IQR, interquartile range; Ref, reference.



Figure 3 | Adjusted marginal means of estimated glomerular filtration rate (eGFR) before and after pregnancy first pregnancy. The green line represents the eGFR of 90 ml/min per 1.73 m². In this model, "age at eGFR measurement" was used as a continuous covariate, and "years before or after pregnancy" was used as a categorical factor. Error bars represent SEM.

in mild AS are excellent with live births reported after 20 weeks' gestation, median gestational age, and mean birth weight comparable to general population data. Second, despite an eGFR step decline after first pregnancy, overall mean eGFR after pregnancy remains >90 ml/min per 1.73 m². Furthermore, pregnancy does not affect eGFR slope after pregnancy. Together, these reassuring results point out that women with mild stage AS should not be

Table 4	Effect of	predictors	on	eGFR	slope	(multilevel
analysis)					

Variables	$\boldsymbol{\beta}$ coefficient	SEM	P value
After first pregnancy	-5.59	2.17	0.010
Age at pregnancy	0.21	0.42	0.617
BMI before pregnancy	0.55	0.40	0.161
Nullipara	-3.78	2.78	0.174
Serum creatinine pre-pregnancy, µmol/L	-0.76	0.10	<0.001
Chronic hypertension	-10.63	5.34	0.047
Proteinuria pre-pregnancy	-13.81	3.55	<0.001
Inheritance pattern			
ADAS	0 (Reference)	-	-
XLAS	-10.14	3.59	0.005
ARAS	-33.42	8.10	<0.001
Glycine substitution	10.60	16.27	0.515
Variant type	NA	NA	0.407
Decennium of pregnancy	NA	NA	0.099

-, not applicable because of reference; ADAS, autosomal dominant Alport syndrome; ARAS, autosomal recessive Alport syndrome; BMI, body mass index; eGFR, estimated glomerular filtration rate; XLAS, X-linked Alport syndrome.

In multilevel analysis, age at eGFR measurement (years) was used as a continuous covariate. All variables were tested in multilevel analysis, adjusted for the effect of time. The dichotomous variable "after first pregnancy" appoints timing of eGFR measurement before/after first pregnancy, with after first pregnancy including the time period of pregnancy interval 1 to 3. Bold parameters are significant (P < 0.05).

dissuaded from becoming pregnant. Third, ARASinheritance pattern shows less favorable pregnancy outcomes. Finally, ARAS-inheritance pattern is an independent predictor for lower birth weight and duration of pregnancy when adjusted for classic risk factors, such as pre-pregnancy kidney function, pre-pregnancy proteinuria, and chronic hypertension, that also show (borderline) significance. The small number of ARAS pregnancies (n =12) should be considered when interpreting this analysis. Furthermore, in this study, *COL4A3/COL4A4* heterozygotes were considered as having ADAS. The significant different outcomes of ARAS pregnancies should be interpreted in the context of comparing them with XLAS pregnancies (66% of cohort) and the relatively mild ADAS pregnancies (24% of cohort).

In this cohort reflecting daily clinical practice of Alport pregnancies, proper quantifiable proteinuria values were missing in 70% of pregnancies, which hinders adjustment in analyses. A bias toward testing after first detection of proteinuria further hinders interpretation of this parameter.

Next to classic risk factors (pre-pregnancy eGFR, proteinuria, and chronic hypertension), different inheritance patterns could be considered when counseling. This study shows the relevance of investigating pregnancy outcomes stratified per cause of kidney disease and the benefits of tailored pre-pregnancy counseling and care for women with CKD.

Furthermore, although the eGFR step decline after first pregnancy was significant, eGFR after second and third pregnancy was not significantly lower. The significant decline

Variable	ALPART cohort CKD G1 (N = 102)	ALPART cohort CKD G2 (N = 23)	Bramham/Wiles CKD G1 (N = 99)	Bramham/Wiles CKD G2 (N = 45)	TOCOS cohort ^a CKD G1 (N = 370)	TOCOS cohort ^a CKD G2 (N = 87)	Dutch general population 2018
Baseline characteristics							
Maternal age, median (IQR), yr Parity, nulliparous Chronic hypertension Proteinuria	29.3 (6.9) 56/96 (58) 12/101 (12) 58/87 (67)	31.1 (4.4) 13/21 (62) 4/19 (21) 18/18 (100)	33.6 (5.3) 36/99 (36) 38/99 (38) 61/99 (62)	33.8 (5.7) 12/45 (27) 26/45 (58) 14/24 (58)	31.3 (5.5) 55 80/370 (22) 0.12 (0–14.6)	33.8 (4.5) 58 36/87 (41) ^b 0.15 (0–6.8) ^b	20 - >40 44 NR NR
Pregnancy outcomes							
Gestational age at delivery, median (IQR) and mean (SD), wk	39.0 (2.39) 38.3 (2.52)	38.0 (3.14) 37.6 (2.71)	38.1 (36.1–40.1) 37.23 (2.98)	37.6 (34.9–40.3) 36.81 (2.46)	37.6 (2.6)	35.7 (3.2)	39.5
Preterm delivery <37 wk <34 wk	18/100 (18) 6/100 (6)	7/23 (30) 4/23 (17)	24/99 (24) 14/99 (14)	16/45 (36) 7/45 (16)	24 7	51 21	7 2
Birth weight, median (IQR), g Gestational hypertension New-onset/doubling proteinuria	3158 (728) 20/91 (22) 35/48 (73)	2669 (688) 6/18 (33) 7/9 (78)	2787 (713) 2/56 (3.6) 42/75 (56)	2625 (627) 1/5 (20) 0/14	2967 (659) 23/290 (8) 76/370 (21)	2484 (707) 9/51 (18) 33/87 (38)	3364 5.3 NR

Table 5 | Comparison to other cohorts for CKD and pregnancy (Piccoli *et al.*, 2015,⁴ Bramham/Wiles 2017–2021⁴⁴ [not all pregnancies in this cohort have been previously published], and Dutch General Population 2018⁴⁵)

ALPART, mAternal and foetal PregnAncy outcomes of women with AlpoRt syndrome; CKD, chronic kidney disease; IQR, interquartile range; NR, not reported; TOCOS, Torino-Cagliari Observational Study.

^aTOCOS cohort included 62% and 43% estimated glomerular filtration rate measurements at first checkup during pregnancy. The classification of CKD categories during pregnancy possibly underestimates CKD category because of glomerular hyperfiltration in the beginning of pregnancy. ^bMeasurement in g/d.

Differences between cohorts on baseline level are: (i) ALPART cohort lower maternal age compared with other cohorts; (ii) ALPART cohort higher incidence nulliparous women; (iii) ALPART cohort lower incidence of chronic hypertension; and (iv) ALPART cohort higher incidence of pre-pregnancy proteinuria compared with UK cohort (TOCOS cohort not comparable). The most important differences in fetal outcome are shown in bold. Differences between cohorts in outcomes are mainly shown in CKD G1 pregnancies: (i) ALPART cohort higher mean gestational age at delivery; (ii) ALPART cohort lower incidence preterm birth; (iii) ALPART cohort higher incidence gestational hypertension during pregnancy. Data are presented as mean (SD) and number/total (percentage) unless stated otherwise. Not all pregnancy outcomes were available for all pregnancies. For categorical variables, incidences are shown as numerator/ denominator.

after first pregnancy might be attributed to almost half of the cohort having only 1 pregnancy. For reasons that could not be retrieved in retrospect (but might be the occurrence of complications during pregnancy or decrease of kidney function), these women have not conceived again. Therefore, the nonsignificant effect of pregnancy on eGFR slope might be overestimated because of "more healthy women" having multiple pregnancies. The lacking validity of the Chronic Kidney Disease Epidemiology Collaboration formula for eGFR values >90 ml/min challenges interpretation of the step decline. Still, because of mean eGFR values remaining >90 ml/min per 1.73 m², clinical significance of this effect is questionable.

Our cohort mainly consisted of women with early CKD (G1–G2) or women with only hematuria without eGFR decline pre-pregnancy, which seems to be the disease category of most *COL4A3-5* women during reproductive age.^{19–21,23,25,26,46,47} Our finding of favorable pregnancy outcomes and long-term kidney outcomes aligns with previous studies.²⁶ These studies also reported remarkable increases of proteinuria during pregnancy. This matches our finding of 76% new onset or doubling of proteinuria, although this parameter is likely skewed because of 60% to 70% missing quantifiable proteinuria measurements. The finding of most proteinuria values restoring to baseline postpartum was likewise skewed.

Baseline characteristics and pregnancy outcomes of CKD G1 to G2 pregnancies in the ALPART cohort, the UK cohort,⁴⁴ the Italian Torino-Cagliari Observational Study (TOCOS) cohort,⁴ and the Dutch general populations are compared in Table 5. At baseline, the ALPART cohort shows lower maternal age, higher nulliparity, and higher prepregnancy proteinuria rates and lower chronic hypertension rates. Overall, neonatal outcomes are most favorable in the G1-ALPART cohort. The longer duration of pregnancy and higher mean birth weight are positive factors for infant neurodevelopment.48-51 A possible explanation for these differences might be the underestimation of CKD categories in the TOCOS cohort, by partly determining CKD categories during pregnancy, and overestimating kidney function attributable to physiological gestational hyperfiltration.^{3,4,7} Furthermore, modifying effects of cause of kidney disease could also explain different outcomes. This specific effect should be further analyzed in future research, correcting not only for CKD category, but also for presence of chronic hypertension and pre-pregnancy proteinuria.

Maternal outcomes, such as new-onset or doubling of proteinuria and gestational hypertension, show higher incidences in the ALPART cohort. Although proteinuria values are likely skewed, as mentioned above, this difference could be partly explained by the pathophysiology of *COL4A3-5*– related disease with leaking glomerular basement membrane and an increased renal blood flow during pregnancy.^{3,6} The higher incidence of gestational hypertension might be attributable to lower incidence of chronic hypertension. Nevertheless, the higher incidence of gestational hypertensive complications during pregnancy, also compared with the Dutch general population, implies suboptimal adaptation to pregnancy in women with *COL4A3-5*–related disease.

Furlano *et al.* recently reported on clinical features of patients with ADAS, of whom 56% were female.¹⁵ They showed an eGFR decline of -1.46 ml/min per 1.73 m² per year. Nothing was reported on pregnancy. Our overall eGFR decline of -1.47 ml/min per 1.73 m², and no significant difference in eGFR slope after pregnancy, matches this finding. Unfortunately, there is no validated formula for eGFR values >90 ml/min. Therefore, translation of absolute decline in eGFR after pregnancy to years of eGFR loss cannot be reliably made.

This is the largest cohort of pregnancies in women with COL4A3-5-related disease. Thanks to an international collaboration with 17 medical centers, we have been able to collect a large cohort and to provide valuable information for the clinician in counseling women with COL4A3-5-related disease who want to conceive. Furthermore, rigorous quality control of the data was undertaken, and much effort was undertaken to obtain information on missing values. In the ALPART cohort, women with a confirmed genetic diagnosis and/or clinical diagnosis were included. Despite >80% of women showing clinical signs of AS pre-pregnancy, some women who received a genetic diagnosis after pregnancy may not have exhibited clinical features of CKD before pregnancy already.³⁵ This information could not be obtained retrospectively. It can be hypothesized that these women with less severe disease have biased results in overestimating our positive results (see Supplementary Tables S5 and S6 for baseline characteristics and pregnancy outcomes stratified on diagnosis before pregnancy). Furthermore, this study can only describe the selection of women with COL4A3-5-related disease within families who presented in the hospital and got tested. Nevertheless, this does not have to be problematic, because study results are applicable to this selection that seeks pre-pregnancy counseling. Furthermore, this cohort is built up via a European Rare Kidney Disease Reference Network survey, and coverage is limited to the knowledge and geographic locations of participants and our international network. Also, inclusion of pregnancies from different countries within a large time span hindered interpretation of outcomes that are influenced by eraand country-dependent clinical management, such as induction of delivery, shortening duration of pregnancy. Nevertheless, a comparison between deliveries before and after 2010 (Supplementary Tables S7 and S8) showed overall similar outcomes. Furthermore, in this international collaboration, differences in international privacy legislation had to be considered. To lower the threshold of participating in our study, only general information on COL4A3-5 genotype was gathered. Therefore, information on base pair alterations was not available, and detailed analyses on genotype-phenotype Our results are highly relevant to clinicians, enabling them to provide personalized pre-pregnancy counseling to patients with *COL4A3-5*–related disease, including stratification on inheritance patterns. Given the reassuring maternal and neonatal results in mild disease stages of AS, there is no reason to dissuade these women from becoming pregnant. Furthermore, it is recommended to optimize disease control before pregnancy in women with CKD. Ideally, patients with AS should be offered pre-pregnancy counseling, opportunity to optimize proteinuria and blood pressure control, consideration of aspirin, and risk stratification for pregnancy care. This also includes being alert to other individuals at risk for AS in their families.

Future research should focus on comparing kidney disease–specific pregnancy outcomes with CKD in general. It is a challenge to find comparable cohorts. Therefore, we have established the PREGeNEPH registry (Pregnancy, Genetics and Nephrology), in which we investigate the influence of specific causes of CKD, next to known risk factors for pregnancy outcomes, such as pre-pregnancy kidney function, chronic hypertension, and proteinuria. In the future, this registry will also gather data prospectively to diminish missing values issues, which are prevalent in current retrospective CKD and pregnancy research.

We conclude that overall neonatal pregnancy outcomes in women with mild COL4A3-5-related disease are favorable. Mean eGFR declines significantly after pregnancy but remains >90 ml/min per 1.73 m². Overall, pregnancy does not significantly affect eGFR slope. ARAS is associated with a higher incidence of preterm birth and is an independent risk factor for lower birth weight, although conclusions should be nuanced because of a small number of ARAS pregnancies and adjustment for pre-pregnancy proteinuria being hindered by missing values. Next to classic risk factors, inheritance pattern could be considered in counseling. With 192 pregnancies from 17 different international centers, this is the largest study to date in this relatively rare disease. This study shows the relevance of investigating pregnancy outcomes in specific kidney disease causes rather than CKD in general.

DISCLOSURE

All the authors declared no competing interests.

DATA STATEMENT

The data supporting the findings of this study are available on submission of a request that will be reviewed by the ALPART working group. Onsite access to the data for verification purposes can be arranged.

FUNDING

This work was supported by the "Stichting Vrienden van het UMC Utrecht," on behalf of the "Elisabeth von Freyburg Stichting" and the "Van Dam-Van Os Stichting" (grant 15121404).

ACKNOWLEDGMENTS

Eleven of the authors of this publication are members of the European Reference Network for Rare Kidney Diseases (ERKNet). We thank the entire ALPART study network for their contribution to this study and the continued collaboration on pregnancy in (rare) kidney disease. Furthermore, we thank the members of the ERKNet Working Group "Hereditary Glomerulopathies" for their contributions and support.

Published abstracts

- Gosselink ME, Snoek R, Van Bakel S, et al. International pregnancy outcomes in Alport syndrome (COL4A3-5 related disease) vs. CKD in general. J Am Soc Nephrol. 2022;33:264.
- Snoek R, Gosselink ME, Vogt L, et al. Primary kidney disease impacts outcome in CKD pregnancies: complications in COL4A3-5 related disease (Alport syndrome) vs. other CKD pregnancies. Nephrol Dial Transplant. 2021;36:gfab131.002.

SUPPLEMENTARY MATERIAL

Supplementary File (Word)

Supplementary Methods. Additional definitions of study parameters.

Supplementary Figure S1. Flowchart of this study.

Supplementary Table S1. Overview of included pregnancies from each participating center.

Supplementary Table S2. Description of complex suspected digenic phenotypes.

Supplementary Table S3. Baseline characteristics of preterm births. **Supplementary Table S4.** Baseline characteristics stratified per inheritance pattern.

Supplementary Table S5. Baseline characteristics stratified on established diagnosis: before pregnancy and after pregnancy.
Supplementary Table S6. Pregnancy outcomes stratified on established diagnosis: before pregnancy and after pregnancy.
Supplementary Table S7. Baseline characteristics stratified on era:

delivery before and after 2010.

Supplementary Table S8. Pregnancy outcomes stratified on era: delivery before and after 2010.

REFERENCES

- Piccoli GB, Zakharova E, Attini R, et al. Pregnancy in chronic kidney disease: need for higher awareness: a pragmatic review focused on what could be improved in the different CKD stages and phases. J Clin Med. 2018;7:415.
- Zhang JJ, Ma XX, Hao L, et al. A systematic review and meta-analysis of outcomes of pregnancy in CKD and CKD outcomes in pregnancy. *Clin J Am Soc Nephrol.* 2015;10:1964–1978.
- Wiles KS, Nelson-Piercy C, Bramham K. Reproductive health and pregnancy in women with chronic kidney disease. *Nat Rev Nephrol.* 2018;14:165–184.
- 4. Piccoli GB, Cabiddu G, Attini R, et al. Risk of adverse pregnancy outcomes in women with CKD. J Am Soc Nephrol. 2015;26:2011–2022.
- van Buren MC, Gosselink M, Groen H, et al. Effect of pregnancy on eGFR after kidney transplantation: a national cohort study. *Transplantation*. 2022;106:1262–1270.
- 6. Wiles K, Webster P, Seed PT, et al. The impact of chronic kidney disease stages 3-5 on pregnancy outcomes. *Nephrol Dial Transplant*. 2021;36: 2008–2017.
- Hladunewich MA, Melamed N, Bramham K. Pregnancy across the spectrum of chronic kidney disease. *Kidney Int*. 2016;89:995–1007.

- Al Khalaf SY, O'Reilly ÉJ, McCarthy FP, et al. Pregnancy outcomes in women with chronic kidney disease and chronic hypertension: a national cohort study. Am J Obstet Gynecol. 2021;225:298.e1–298.e20.
- 9. Snoek R, van der Graaf R, Meinderts JR, et al. Pregnancy in advanced kidney disease: clinical practice considerations on a challenging combination. *Nephron*. 2020;144:185–189.
- **10.** Kendrick J, Sharma S, Holmen J, et al. Kidney disease and maternal and fetal outcomes in pregnancy. *Am J Kidney Dis.* 2015;66:55–59.
- 11. Savige J, Colville D, Rheault M, et al. Alport syndrome in women and girls. *Clin J Am Soc Nephrol*. 2016;11:1713–1720.
- Kashtan CE, Ding J, Garosi G, et al. Alport syndrome: a unified classification of genetic disorders of collagen IV α345: a position paper of the Alport Syndrome Classification Working Group. *Kidney Int.* 2018;93: 1045–1051.
- 13. Mastrangelo A, Giani M, Groppali E, et al. X-linked alport syndrome in women: genotype and clinical course in 24 cases. *Front Med (Lausanne)*. 2020;7:580376.
- 14. Kashtan CE. Alport syndrome. In: Adam MP, Mirzaa GM, Pagon RA, et al, eds. GeneReviews.University of Washington, Seattle. Copyright 1993–2023.
- Furlano M, Martínez V, Pybus M, et al. Clinical and genetic features of autosomal dominant Alport syndrome: a cohort study. Am J Kidney Dis. 2021;78:560–570.e561.
- 16. Torra R, Furlano M. New therapeutic options for Alport syndrome. *Nephrol Dial Transplant*. 2019;34:1272–1279.
- Savige J, Lipska-Zietkiewicz BS, Watson E, et al. Guidelines for genetic testing and management of Alport syndrome. *Clin J Am Soc Nephrol*. 2022;17:143–154.
- Gibson J, Fieldhouse R, Chan MMY, et al. Prevalence estimates of predicted pathogenic COL4A3-COL4A5 variants in a population sequencing database and their implications for Alport syndrome. J Am Soc Nephrol. 2021;32:2273–2290.
- Matsubara S, Muto S. Good obstetric outcome of consecutive pregnancies in a woman with Alport syndrome. *Arch Gynecol Obstet*. 2012;286:261–262.
- 20. Matsubara S, Ueda Y, Takahashi H, et al. Pregnancy complicated with Alport syndrome: a good obstetric outcome and failure to diagnose an infant born to a mother with Alport syndrome by umbilical cord immunofluorescence staining. *J Obstet Gynaecol Res.* 2009;35: 1109–1114.
- 21. Crovetto F, Moroni G, Zaina B, et al. Pregnancy in women with Alport syndrome. *Int Urol Nephrol.* 2013;45:1223–1227.
- 22. Mehta S, Saifan C, Abdellah M, et al. Alport's syndrome in pregnancy. *Case Rep Med.* 2013;2013:374020.
- 23. Alessi M, Fabris A, Zambon A, et al. Pregnancy in Alport syndrome: a report of two differently-evolving cases. J Obstet Gynaecol. 2014;34:98–100.
- 24. Kitanovska BG, Gerasimovska V, Livrinova V. Two pregnancies with a different outcome in a patient with Alport syndrome. *Open Access Maced J Med Sci.* 2016;4:439–442.
- 25. Nishizawa Y, Takei T, Miyaoka T, et al. Alport syndrome and pregnancy: good obstetric and nephrological outcomes in a pregnant woman with homozygous autosomal recessive Alport syndrome. *J Obstet Gynaecol Res.* 2016;42:331–335.
- 26. Yefet E, Tovbin D, Nachum Z. Pregnancy outcomes in patients with Alport syndrome. *Arch Gynecol Obstet*. 2016;293:739–747.
- Drury ER, Stillman IE, Pollak MR, Denker BM. Autosomal recessive Alport syndrome unveiled by pregnancy. *Nephron*. 2019;143:288–292.
- 28. Pepe F, Di Guardo F, Zambrotta E, et al. Renal impairment in Alport syndrome pregnant woman: case report and review of the literature. *Clin Case Rep.* 2020;8:3003–3007.
- 29. Jehn U, Müller-Hofstede C, Heitplatz B, et al. A novel homozygous mutation in the COL4A4 gene (Gly1436del) causing Alport syndrome exposed by pregnancy: a case report and review of the literature. *Case Rep Nephrol.* 2022;2022:5243137.
- **30.** Richards S, Aziz N, Bale S, et al. Standards and guidelines for the interpretation of sequence variants: a joint consensus recommendation of the American College of Medical Genetics and Genomics and the Association for Molecular Pathology. *Genet Med.* 2015;17:405–424.
- 31. Castor EDC. Castor electronic data capture. Published 2019. Accessed August 28, 2019. https://castoredc.com
- **32.** Inker LA, Eneanya ND, Coresh J, et al. New creatinine- and cystatin Cbased equations to estimate GFR without race. *N Engl J Med.* 2021;385: 1737–1749.

- **33.** Webster K, Fishburn S, Maresh M, et al. Diagnosis and management of hypertension in pregnancy: summary of updated NICE guidance. *BMJ*. 2019;366:15119.
- **34.** Macdonald-Wallis C, Silverwood RJ, Fraser A, et al. Gestational-agespecific reference ranges for blood pressure in pregnancy: findings from a prospective cohort. *J Hypertens*. 2015;33:96–105.
- Levin A, Stevens PE. Summary of KDIGO 2012 CKD Guideline: behind the scenes, need for guidance, and a framework for moving forward. *Kidney Int.* 2014;85:49–61.
- **36.** Levey AS, Eckardt KU, Dorman NM, et al. Nomenclature for kidney function and disease: report of a Kidney Disease: Improving Global Outcomes (KDIGO) Consensus Conference. *Kidney Int.* 2020;97:1117–1129.
- Kiserud T, Piaggio G, Carroli G, et al. The World Health Organization fetal growth charts: a multinational longitudinal study of ultrasound biometric measurements and estimated fetal weight. *PLoS Med*. 2017;14:e1002220.
- **38.** Gosselink ME, van Buren MC, Kooiman J, et al. A nationwide Dutch cohort study shows relatively good pregnancy outcomes after kidney transplantation and finds risk factors for adverse outcomes. *Kidney Int.* 2022;102:866–875.
- **39.** Shou H, Hsu JY, Xie D, et al. Analytic considerations for repeated measures of eGFR in cohort studies of CKD. *Clin J Am Soc Nephrol*. 2017;12:1357–1365.
- 40. Matsushita K, van der Velde M, Astor BC, et al. Association of estimated glomerular filtration rate and albuminuria with all-cause and cardiovascular mortality in general population cohorts: a collaborative meta-analysis. *Lancet*. 2010;375:2073–2081.
- **41.** Gansevoort RT, Matsushita K, van der Velde M, et al. Lower estimated GFR and higher albuminuria are associated with adverse kidney outcomes: a collaborative meta-analysis of general and high-risk population cohorts. *Kidney Int.* 2011;80:93–104.

- 42. Remuzzi G, Benigni A, Remuzzi A. Mechanisms of progression and regression of renal lesions of chronic nephropathies and diabetes. *J Clin Invest*. 2006;116:288–296.
- **43.** Halbesma N, Jansen DF, Heymans MW, et al. Development and validation of a general population renal risk score. *Clin J Am Soc Nephrol.* 2011;6:1731–1738.
- 44. Bramham K, Seed PT, Lightstone L, et al. Diagnostic and predictive biomarkers for pre-eclampsia in patients with established hypertension and chronic kidney disease. *Kidney Int.* 2016;89:874–885.
- 45. Groenendaal F, Kwee A, de Miranda E, et al. Perinatale zorg in Nederland anno 2018. Landelijke perinatale cijfers en duiding. Published 2018. Accessed December 15, 2022. https://assets.perined.nl/docs/fc23b860-a5 ff-4ef6-b164-aedf7881cbe3.pdf
- Brunini F, Zaina B, Gianfreda D, et al. Alport syndrome and pregnancy: a case series and literature review. Arch Gynecol Obstet. 2018;297:1421–1431.
- 47. Matsuo K, Tudor EL, Baschat AA. Alport syndrome and pregnancy. *Obstet Gynecol.* 2007;109(pt2):531–532.
- 48. Zwertbroek EF, Franssen MTM, Broekhuijsen K, et al. Neonatal developmental and behavioral outcomes of immediate delivery versus expectant monitoring in mild hypertensive disorders of pregnancy: 2year outcomes of the HYPITAT-II trial. Am J Obstet Gynecol. 2019;221:154. e1–154.e11.
- **49.** Richards M, Hardy R, Kuh D, Wadsworth ME. Birth weight and cognitive function in the British 1946 birth cohort: longitudinal population based study. *BMJ*. 2001;322:199–203.
- 50. Matte TD, Bresnahan M, Begg MD, Susser E. Influence of variation in birth weight within normal range and within sibships on IQ at age 7 years: cohort study. *BMJ*. 2001;323:310–314.
- 51. Sørensen HT, Sabroe S, Olsen J, et al. Birth weight and cognitive function in young adult life: historical cohort study. *BMJ*. 1997;315: 401–403.